Author’s response to reviews

Title: Mapping oral health related quality of life to generic health state values

Authors:

David S Brennan (david.brennan@adelaide.edu.au)
A. John Spencer (john.spencer@adelaide.edu.au)

Version: 3 Date: 4 August 2006

Author’s response to reviews:

We submit a final formatted version of a manuscript titled 'Mapping oral health related quality of life to generic health state values' by Brennan and Spencer that has been accepted for publication as a scientific article in BMC Health Services Research. As advised, we have removed the keywords section, written that the authors have no competing interests, and listing the authors' contributions.

Yours sincerely
David S. Brennan
A. John Spencer
Australian Research Centre for Population Oral Health, The University of Adelaide